\begin{thebibliography}{10}

\bibitem{supplementaryResults}
Supplementary tables:
  http://unbox.org/wisp/var/ekrem/kernel/supplementaryresults/.

\bibitem{Alpaydin2004}
E.~Alpaydin.
\newblock {\em Introduction to Machine Learning}.
\newblock MIT Press, 2004.

\bibitem{Angelis2000}
L.~Angelis and I.~Stamelos.
\newblock A simulation tool for efficient analogy based cost estimation.
\newblock {\em Empirical Softw. Engg.}, 5(1):35--68, 2000.

\bibitem{Auer2006}
M.~Auer, A.~Trendowicz, B.~Graser, E.~Haunschmid, and S.~Biffl.
\newblock Optimal project feature weights in analogy-based cost estimation:
  Improvement and limitations.
\newblock {\em IEEE Trans. Softw. Eng.}, 32:83--92, 2006.

\bibitem{baker07}
D.~Baker.
\newblock A hybrid approach to expert and model-based effort estimation.
\newblock Master's thesis, Lane Department of Computer Science and Electrical
  Engineering, West Virginia University, 2007.

\bibitem{Boehm2000}
B.~Boehm, C.~Abts, and S.~Chulani.
\newblock Software development cost estimation approaches: A survey.
\newblock {\em Annals of Software Engineering}, 10:177--205, 2000.

\bibitem{Boehm1981}
B.~W. Boehm.
\newblock {\em Software Engineering Economics}.
\newblock Prentice Hall PTR, Upper Saddle River, NJ, USA, 1981.

\bibitem{Briand1999}
L.~C. Briand, K.~El~Emam, D.~Surmann, I.~Wieczorek, and K.~D. Maxwell.
\newblock An assessment and comparison of common software cost estimation
  modeling techniques.
\newblock pages 313--322, 1999.

\bibitem{Conte1986}
S.~D. Conte, H.~E. Dunsmore, and V.~Y. Shen.
\newblock {\em Software engineering metrics and models}.
\newblock Benjamin-Cummings Publishing Co., Inc., Redwood City, CA, USA, 1986.

\bibitem{Cressie1993}
N.~A.~C. Cressie.
\newblock {\em Statistics for Spatial Data (Wiley Series in Probability and
  Statistics)}.
\newblock Wiley-Interscience, 1993.

\bibitem{Desharnais1989}
J.~Desharnais.
\newblock Analyse statistique de la productivitie des projets informatique a
  partie de la technique des point des fonction.
\newblock Master's thesis, Univ. of Montreal, 1989.

\bibitem{Foss}
T.~Foss, E.~Stensrud, B.~Kitchenham, and I.~Myrtveit.
\newblock A simulation study of the model evaluation criterion mmre.
\newblock {\em IEEE Trans. Softw. Eng.}, vol:29no11pp985--995, 2003.

\bibitem{Frank03}
E.~Frank, M.~Hall, and B.~Pfahringer.
\newblock Locally weighted naive bayes.
\newblock In {\em Proceedings of the Conference on Uncertainty in Artificial
  Intelligence}, pages 249--256. Morgan Kaufmann, 2003.

\bibitem{jeffery2001}
R.~Jeffery, M.~Ruhe, and I.~Wieczorek.
\newblock Using public domain metrics to estimate software development effort.
\newblock In {\em METRICS '01: Proceedings of the 7th International Symposium
  on Software Metrics}, page~16, Washington, DC, USA, 2001. IEEE Computer
  Society.

\bibitem{John1995}
G.~John and P.~Langley.
\newblock Estimating continuous distributions in bayesian classifiers.
\newblock In {\em In Proceedings of the Eleventh Conference on Uncertainty in
  Artificial Intelligence}, pages 338--345. Morgan Kaufmann, 1995.

\bibitem{Jorgensen2004}
M.~Jorgensen.
\newblock {A review of studies on expert estimation of software development
  effort}.
\newblock {\em Journal of Systems and Software}, 70:37--60, February 2004.

\bibitem{Jorgensen2009}
M.~Jorgensen and T.~Gruschke.
\newblock The impact of lessons-learned sessions on effort estimation and
  uncertainty assessments.
\newblock {\em IEEE Trans. Softw. Eng.}, 35(3):368--383, May-June 2009.

\bibitem{Jorgensen2007}
M.~Jorgensen and M.~Shepperd.
\newblock A systematic review of software development cost estimation studies.
\newblock {\em IEEE Trans. Softw. Eng.}, 33(1):33--53, 2007.

\bibitem{Kadoda2000}
G.~Kadoda, M.~Cartwright, and M.~Shepperd.
\newblock {On configuring a case-based reasoning software project prediction
  system}.
\newblock {\em UK CBR Workshop, Cambridge, UK}, pages 1--10, 2000.

\bibitem{kemerer87}
C.~Kemerer.
\newblock An empirical validation of software cost estimation models.
\newblock {\em Communications of the ACM}, 30(5):416--429, May 1987.

\bibitem{Kitchenham2009}
B.~Kitchenham and E.~Mendes.
\newblock Why comparative effort prediction studies may be invalid.
\newblock In {\em PROMISE '09: Proceedings of the 5th International Conference
  on Predictor Models in Software Engineering}, pages 1--5, New York, NY, USA,
  2009. ACM.

\bibitem{Mendes2007}
B.~Kitchenham, E.~Mendes, and G.~H. Travassos.
\newblock Cross versus within-company cost estimation studies: A systematic
  review.
\newblock {\em IEEE Trans. Softw. Eng.}, 33(5):316--329, 2007.
\newblock Member-Kitchenham, Barbara A.

\bibitem{Kitchenham2007}
B.~A. Kitchenham, E.~Mendes, and G.~H. Travassos.
\newblock Cross versus within-company cost estimation studies: A systematic
  review.
\newblock {\em IEEE Trans. Softw. Eng.}, 33(5):316--329, 2007.

\bibitem{Li2006}
J.~Li and G.~Ruhe.
\newblock {A comparative study of attribute weighting heuristics for effort
  estimation by analogy}.
\newblock {\em Proceedings of the 2006 ACM/IEEE international symposium on
  Empirical software engineering}, page~74, 2006.

\bibitem{Li2009}
Y.~Li, M.~Xie, and T.~Goh.
\newblock {A study of project selection and feature weighting for analogy based
  software cost estimation}.
\newblock {\em Journal of Systems and Software}, 82:241--252, 2009.

\bibitem{Lum2008}
K.~Lum, T.~Menzies, and D.~Baker.
\newblock {2CEE, A TWENTY FIRST CENTURY EFFORT ESTIMATION METHODOLOGY}.
\newblock {\em ISPA / SCEA}, pages 12 -- 14, 2008.

\bibitem{Mendes2002a}
E.~Mendes and N.~Mosley.
\newblock Further investigation into the use of cbr and stepwise regression to
  predict development effort for web hypermedia applications.
\newblock {\em Empirical Software Engineering, International Symposium on},
  0:79, 2002.

\bibitem{Mendes2008a}
E.~Mendes and N.~Mosley.
\newblock Bayesian network models for web effort prediction: A comparative
  study.
\newblock {\em IEEE Trans. Softw. Eng.}, 34:723--737, 2008.

\bibitem{Mendes2002}
E.~Mendes, N.~Mosley, and I.~Watson.
\newblock A comparison of case-based reasoning approaches.
\newblock In {\em WWW '02: Proceedings of the 11th international conference on
  World Wide Web}, pages 272--280, New York, NY, USA, 2002. ACM.

\bibitem{Mendes2003}
E.~Mendes, I.~D. Watson, C.~Triggs, N.~Mosley, and S.~Counsell.
\newblock A comparative study of cost estimation models for web hypermedia
  applications.
\newblock {\em Empirical Software Engineering}, 8(2):163--196, 2003.

\bibitem{Menzies2006}
T.~Menzies, Z.~Chen, J.~Hihn, and K.~Lum.
\newblock Selecting best practices for effort estimation.
\newblock {\em IEEE Trans. Softw. Eng.}, 32:883--895, 2006.

\bibitem{Milic2004}
D.~Milic and C.~Wohlin.
\newblock {Distribution Patterns of Effort Estimations}.
\newblock In {\em Euromicro}, 2004.

\bibitem{Molokken-Ostvold2004}
K.~Mol{\o}kken-{\O}stvold, M.~J{\o}rgensen, S.~S. Tanilkan, H.~Gallis, A.~C.
  Lien, and S.~E. Hove.
\newblock A survey on software estimation in the norwegian industry.
\newblock {\em IEEE International Symposium on Software Metrics}, pages
  208--219, 2004.

\bibitem{Myrtveit1999}
I.~Myrtveit and E.~Stensrud.
\newblock A controlled experiment to assess the benefits of estimating with
  analogy and regression models.
\newblock {\em IEEE Trans. Softw. Eng.}, 25:510--525, 1999.

\bibitem{Myrtveit}
I.~Myrtveit, E.~Stensrud, and M.~Shepperd.
\newblock Reliability and validity in comparative studies of software
  prediction models.
\newblock {\em IEEE Trans. Softw. Eng.}, vol:31no5pp380--391, May 2005.

\bibitem{Palpanas2003}
T.~Palpanas, D.~Papadopoulos, V.~Kalogeraki, and D.~Gunopulos.
\newblock Distributed deviation detection in sensor networks.
\newblock {\em SIGMOD Rec}, 32:2003, 2003.

\bibitem{Pendharkar2005}
P.~C. Pendharkar, G.~H. Subramanian, and J.~A. Rodger.
\newblock A probabilistic model for predicting software development effort.
\newblock {\em IEEE Trans. Softw. Eng.}, 31:615--624, 2005.

\bibitem{Premraj2007}
R.~Premraj and T.~Zimmermann.
\newblock {Building Software Cost Estimation Models using Homogenous Data}.
\newblock {\em ESEM}, 2007.

\bibitem{Robson2002}
C.~Robson.
\newblock {Real world research: a resource for social scientists and
  practitioner-researchers}.
\newblock {\em Blackwell Publisher Ltd}, 2002.

\bibitem{Scheid2004}
S.~Scheid.
\newblock Introduction to kernel smoothing.
\newblock Talk, January 2004.

\bibitem{Scott1992}
D.~W. Scott.
\newblock {\em Multivariate Density Estimation: Theory, Practice, and
  Visualization (Wiley Series in Probability and Statistics)}.
\newblock Wiley-Interscience, September 1992.

\bibitem{Shepperd2007}
M.~Shepperd.
\newblock Software project economics: a roadmap.
\newblock In {\em FOSE '07: 2007 Future of Software Engineering}, pages
  304--315, 2007.

\bibitem{Shepperd2001}
M.~Shepperd and M.~Cartwright.
\newblock Predicting with sparse data.
\newblock {\em IEEE Trans. Softw. Eng.}, 27:987--998, 2001.

\bibitem{Kadoda2001}
M.~Shepperd and G.~Kadoda.
\newblock Comparing software prediction models using simulation.
\newblock {\em IEEE Trans. Softw. Eng.}, pages 1014--1022, 2001.

\bibitem{Shepperd1997}
M.~Shepperd and C.~Schofield.
\newblock Estimating software project effort using analogies.
\newblock {\em IEEE Trans. Softw. Eng.}, 23(11):736--743, 1997.

\bibitem{Shepperd1996}
M.~Shepperd, C.~Schofield, and B.~Kitchenham.
\newblock Effort estimation using analogy.
\newblock In {\em International Conference on Software Engineering}, pages
  170--178, 1996.

\bibitem{Stensrud2002}
E.~Stensrud, T.~Foss, B.~Kitchenham, and I.~Myrtveit.
\newblock An empirical validation of the relationship between the magnitude of
  relative error and project size.
\newblock In {\em METRICS '02: Proceedings of the 8th International Symposium
  on Software Metrics}, page~3, Washington, DC, USA, 2002. IEEE Computer
  Society.

\bibitem{Stensrud}
E.~Stensrud, T.~Foss, B.~Kitchenham, and I.~Myrtveit.
\newblock A further empirical investigation of the relationship between mre and
  project size.
\newblock {\em Empirical Software Engineering}, 8(2):139--161, 2003.

\bibitem{Wand1994}
M.~P. Wand and M.~C. Jones.
\newblock {\em Kernel Smoothing (Monographs on Statistics and Applied
  Probability)}.
\newblock {Chapman \& Hall/CRC}, December 1994.

\end{thebibliography}

\section{Results}
\label{sec:results}

To see the effect of kernel weighting, rigorous experimentation on 19 datasets and 3 different performance measures was performed.
Also, for each performance measure we tried 4 different kernels subject to 5 different bandwidths, plus the IRWM kernel (which does not have a bandwidth concept) and reported associated $win,tie,loss$ statistics.

\input{sampleTable}

\fig{win-tie-loss-desharnais} is a sample table, where we report the $win,tie,loss$ 
statistics of Desharnais dataset for ABE0 and N-ABE through Gaussian kernel. 
For each dataset we have $4$ such tables (one for each kernel), so for all datasets there is \textit{19 Datasets $\times$ 4 tables = 76 tables}.
For the IRWM kernel there will be another \textit{19 datasets $\times$ 1 kernel = 19 tables}.
So in total, we have \textit{76 + 19 = 95 tables} to report, which is impractical to put in a summary report such as this one.
Therefore, we provide the sample table of \fig{win-tie-loss-desharnais} as a pointer and manual to other 94 tables, which are given under \cite{supplementaryResults}.

In \fig{win-tie-loss-desharnais} we see that each row reports $win,tie,loss$ statistics of ABE0 methods (\textit{k=3,5,7,9,best}) as well as N-ABE methods (\textit{k=[3,5,7,9,best]+kern where kern stands for kernel weighting}) subject to a particular performance measure.
Similarly, each column shows the $win,tie,loss$ statistics associated with a particular bandwidth value.
As can be seen in \fig{win-tie-loss-desharnais}, ABE0 methods always have higher $win$ values and always have a $loss$ value of $0$, meaning that they never lose against N-ABE methods.
By analyzing each row/column intersection we can see that the performance of ABE0 is never improved by N-ABE.

The analysis of \fig{win-tie-loss-desharnais} is only valid for Desharnais dataset and Gaussian kernel.
So as to provide an executive summary of all 19 datasets and all kernel/bandwidth combinations, we will use \fig{summary-all}.
Each row of \fig{summary-all} shows the comparison of ABE0 performance to that of N-ABE subject to 3 different performance measures.
Every kernel/bandwidth intersection in \fig{summary-all} has 3 symbols corresponding to MdMRE, MAR and Pred(25) comparisons from left to right.
Each of these 3 symbols can have 3 values: $-,+,o$.
A \textit{``$-$''} symbol means that N-ABE decreased the accuracy of ABE0, whereas a \textit{``$o$''} symbol means ABE0 and N-ABE are statistically same in all conditions and a \textit{``$+$''} symbol shows that ABE0 accuracy was improved through kernel weighting (i.e. N-ABE has a better accuracy than ABE0).

So as to decide if ABE0 performance is improved ($+$), decreased ($-$) or not changed ($o$) by N-ABE we look at all of the $95$ detail-tables such \fig{win-tie-loss-desharnais} and see the behavior of 5 \textit{k} values under ABE0 and N-ABE.
If the majority of \textit{k} values (at least 3 out of 5) are improved by N-ABE in terms of $win-loss$ values, then we assign a ``$+$'' symbol for that setting; if the performance associated with the majority of \textit{k} values is decreased by N-ABE, then we assign a ``$-$'' symbol; otherwise, we assign a ``$o$'' symbol to that setting.

See in \fig{summary-all} that out of $19$ datasets, there is only one dataset (SDR) where N-ABE provides a performance improvement in certain cases. 
Even for that dataset there are $15$ ``$+$'' symbols and $21$ ``$-$'' symbols, meaning that most of the time N-ABE is still destructive.
In $18$ datasets, which is $\frac{18}{19} = 95\%$ of all the datasets, there is not a single case where N-ABE improves the performance of ABE0.
Also note that \fig{summary-all} provides an extensive summary of different error measures (MdMRE, MAR, Pred(25)) as well as kernels and bandwidths.
Therefore, our conclusion from \fig{summary-all} that \textit{``non-uniform weighting through standard kernel methods does not improve the performance of ABE''} is quite stable accross different datasets and error measures.

\input{allDatasetResults}

\input{irwm-results}

Another summary table is given in \fig{summary-irwm}.
\fig{summary-irwm} is very similar to \fig{summary-all} in the sense that it summarizes the performance of N-ABE over 19 datasets w.r.t. three different performance measures.
The difference is that \fig{summary-all} summarizes the results of standard kernel methods, whereas in \fig{summary-irwm} we see the N-ABE performance under an \textit{expert-based} kernel, i.e. IRWM.
Although there are important differences between standard and expert-based kernels (IRWM has no bandwidth parameter), the results seen in \fig{summary-irwm} is quite similar to those of \fig{summary-all}.
As can be seen in \fig{summary-irwm}, there is not a single case where N-ABE (under IRWM kernel) improves the performance of ABE0.
Furthermore, the amount of ``$-$'' symbols is much more than ``$o$'', meaning that N-ABE decreases the performance of ABE0 most of the time.

%As we mentioned earlier, we will evaluate the effect of non-uniform weighting against uniform weighting in an ABE context according to three performance measures: MdMRE, Pred(25) and Win-tie-loss values.
%In this section we present our results.
%We first evaluate MAR, MdMRE and Pred(25) results for each dataset and then present related win-tie-loss values. 
%In this paper, for standard kernel methods only Cocomo81 results are included due to 2 reasons: 1) We have strict page restrictions  and result tables consume too much space, 2) all observations from Cocomo81 are valid for other datasets, i.e. non-uniform weighting never improves standad ABE.
%For not-included tables please contact the authors or see \cite{supplementaryResults}.
 
%\subsection{Evaluation of MAR, MdMRE and Pred(25) Results}
%\label{subsec:mre-based-results}
%
%%The MRE-based measures we investigate in this section are MdMRE and MMRE.
%The MAR, MdMRE and Pred(25) values for Cocomo81 is provided in Figure \ref{fig:mre-coc81}.
%In this Figure N-ABE settings are indicated with a $+N$ sign, where $+N$ stands for \textbf{\textit{N}}on-uniform weighting.
%Furthermore, different number of analogies used in each treatment are represented by actual \textit{k} values, whereas \textit{\textbf{d}ynamic k} is represented by a $d$ symbol.
%
%
%\subsubsection{Results for Cocomo81}
%In Figure \ref{fig:mre-coc81} we see MAR, MdMRE and Pred(25) values for Cocomo81.
%The first observation we can make out of Figure \ref{fig:mre-coc81} is that for the same \textit{k} value N-ABE methods fail to improve ABE0, i.e. kernel methods cannot provide a performance improvement.
%%For all settings, uniform weighting yields lower error rates.
%Application of different kernels as well as different bandwidths for non-uniform weighting does not produce a recognizable pattern in terms of MAR, MdMRE and Pred(25).
%%Change of bandwidths for kernels does not create a recognizable pattern in the results either.
%Another observation from Figure \ref{fig:mre-coc81} is that both for uniform and non-uniform weighting, smaller \textit{k} values yield lower MdMRE and Pred(25) values.
%%Regarding Figure \ref{fig:mre-coc81}, note that we use all $63$ instances of Cocomo81 dataset with full range of its features in all treatments.
%
%
%
%
%\subsubsection{IRWM Results for All Datasets}
%Figure \ref{fig:irwm-mdmre-mmre} presents MAR, MdMRE and Pred(25) results for IRWM kernel.
%The difference of results in Figure \ref{fig:irwm-mdmre-mmre} is that previous results of N-ABE methods came from standard kernel methods (minimum human interaction), whereas in Figure \ref{fig:irwm-mdmre-mmre}  weighting strategy is defined by human experts (complete human dependence).
%Despite the difference in the weighting strategies, the trend in the results are very alike, i.e. in none of the 3 datasets can N-ABE methods outperform ABE0.
%%Therefore, after 330 settings which involve both non-parametric methods and expert methods for weighting N-ABE, we still do not observe a case where N-ABE methods could bring an improvement on simple ABE0 method.
%
%
%
%\subsection{Evaluation of WIN-TIE-LOSS Results}
%
%%In this section we evaluate the performance of non-uniform weighting and uniform weighting according to  win-tie-loss values.
%%Since we have 10 settings for each kernel subject to 20 runs, the sum of $win$, $tie$ and $loss$ values can be at most 180 (\textit{(10 settings - 1 setting itself) * 20 runs = 180}).
%%The nice point of win-tie-loss calculation is that it does not only show us whether N-ABE with kernel density estimation provides an improvement to ABE0, but it also shows us the performance of a single method (ABE0 or N-ABE) in comparison to all other methods.
%
%\subsubsection{Results for Cocomo81}
%In Figure \ref{fig:win-tie-loss-coc81} the win-tie-loss values for Cocomo81 are given.
%Since we have 10 settings for each kernel subject to 20 runs, the sum of $win$, $tie$ and $loss$ values can be at most 180 (\textit{(10 settings - 1 setting itself) * 20 runs = 180}).
%For all settings, the $tie$ values are most of the time less than 45 (less than 25\% of all the comparisons), which means that in 75\% or more of the comparisons there is a statistical difference between the methods.
%%The first observation we can make is that smaller number of analogies have always attained higher $win$ values and lower $loss$ values.
%However, when ABE0 is compared to N-ABE methods for different \textit{k} values, N-ABE can never improve the $win$ values over ABE0.
%
%From Figure \ref{fig:win-tie-loss-coc81} we can also see the effect of applying different kernels and different bandwidths on the performance of N-ABE.
%In terms of kernels, there is not a considerable performance difference between different types.
%This is consistent with prior research, which reports that different kernels yield similar results~\cite{Cressie1993}.
%%For Cocomo81 dataset we observe that the same fact is also valid for software effort estimation data. 
%On the other hand, our results in terms of bandwidth is contradictory.
%Bandwidth is reported to be influential in different contexts~\cite{Cressie1993, Scott1992, Wand1994}.
%However, we are unable to observe a considerable effectdue to different bandwidths.
%%In Figure \ref{fig:win-tie-loss-coc81} the win-tie-loss values of different kernels when used with 5 different bandwidths are very similar.
%%In fact, for the uniform kernel the performance is completely identical between different bandwidths.
%Therefore, 
%%we can propose that software effort data behaves differently when compared to other data types, i.e. 
%unlike the performance on spatial data performance on software effort data is not improved by changes in bandwidth. 
%
%
%
%\subsubsection{IRWM Results for All Datasets}
%%Up to this point we have observed 315 different settings and saw that neither kernel nor the bandwidth change does have a considerable impact on the performance of N-ABE.
%%Furthermore, we found out that simple ABE0 approach yields higher performance measures in terms of win-tie-loss values.
%%However, standard kernels are not the only alternatives of non-uniform weighting.
%%Another N-ABE methods we use in this research is so called IRWM~\cite{Mendes2002, Mendes2003}.
%The win-tie-loss values of all $3$ datasets for IRWM weighted N-ABE are given in Figure \ref{fig:irwm-win-tie-loss}.
%Since IRWM is a different weighting approach than kernel density estimation, we do not have bandwidths to compare in that scenario.
%%On the other hand with IRWM results we can mutually compare the estimation performances of N-ABE and ABE0 approaches.
%Our reading from Figure \ref{fig:irwm-win-tie-loss} is that N-ABE methods cannot outperform ABE0 in any one of the three datasets.
%In other words, just like standard kernel weighted N-ABE, IRWM weighted N-ABE also fails to outperform ABE0.
%Therefore, in a total of 330 settings we see that N-ABE methods are unable to outperform the performance of simple ABE0 approach.

